

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Matousek, Jindrich

Working Paper Individual discount rates: A meta-analysis of the experimental evidence

IES Working Paper, No. 40/2018

Provided in Cooperation with: Charles University, Institute of Economic Studies (IES)

Suggested Citation: Matousek, Jindrich (2018) : Individual discount rates: A meta-analysis of the experimental evidence, IES Working Paper, No. 40/2018, Charles University in Prague, Institute of Economic Studies (IES), Prague

This Version is available at: https://hdl.handle.net/10419/203219

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Institute of Economic Studies, Faculty of Social Sciences Charles University in Prague

Individual Discount Rates: A Meta-Analysis of the Experimental Evidence

Jindrich Matousek

IES Working Paper: 40/2018



Institute of Economic Studies, Faculty of Social Sciences, Charles University in Prague

[UK FSV – IES]

Opletalova 26 CZ-110 00, Prague E-mail : <u>ies@fsv.cuni.cz</u> <u>http://ies.fsv.cuni.cz</u>

Institut ekonomických studií Fakulta sociálních věd Univerzita Karlova v Praze

> Opletalova 26 110 00 Praha 1

E-mail : <u>ies@fsv.cuni.cz</u> <u>http://ies.fsv.cuni.cz</u>

Disclaimer: The IES Working Papers is an online paper series for works by the faculty and students of the Institute of Economic Studies, Faculty of Social Sciences, Charles University in Prague, Czech Republic. The papers are peer reviewed. The views expressed in documents served by this site do not reflect the views of the IES or any other Charles University Department. They are the sole property of the respective authors. Additional info at: <u>ies@fsv.cuni.cz</u>

Copyright Notice: Although all documents published by the IES are provided without charge, they are licensed for personal, academic or educational use. All rights are reserved by the authors.

Citations: All references to documents served by this site must be appropriately cited.

Bibliographic information:

Matousek, J. (2018): "Individual Discount Rates: A Meta-Analysis of the Experimental Evidence". IES Working Papers 40/2018, IES FSV, Charles University.

This paper can be downloaded at: http://ies.fsv.cuni.cz

Individual Discount Rates: A Meta-Analysis of the Experimental Evidence

Jindrich Matousek^a

^aInstitute of Economic Studies, Faculty of Social Sciences, Charles University Opletalova 21, 110 00, Prague, Czech Republic Email (corresponding author): <u>jindrich.matousek@fsv.cuni.cz</u>

December 2018

Abstract:

We examine 434 estimates of the individual discount rate reported in 27 published studies. The estimates vary substantially across studies with reported mean at the value of 0.4. We detect presence of selective reporting in the discounting literature using a meta-analytical methods. Our results suggest that relevant discounting literature overestimates the discount rate approximately twofold. We apply Bayesian model averaging to explain heterogeneity in the estimates. Discount rate estimates are influenced mainly by the experimental design used for its elicitation. We confirm domain independence and the effect of time horizon presented within the discounting task. Our results support the external validity of experiments conducted on student samples and the utilisation of hypothetical rewards in experiments.

JEL Classification: D01, C83, C90

Keywords: Discount rate, Discounting, Experiment, Publication bias, Meta-Analysis, Bayesian Model Averaging

Acknowledgements: This research was supported by the Grant Agency of Charles University (grant no. 848217) and the SVV project no. 260463.

1 Introduction

Most important decisions on the private as well as public level often involve intertemporal choices and tradeoffs as decisions of today influence the future. Correct evaluation of future impacts of those choices on our well-being is crucial. Economic decisions intervene into various individual domains such as health, education, savings or business, but also into public domains such as government interventions, policies and social projects. Welfare analyses often involve non-monetary aspects for which a social discount rate is more applicable than a standard financial market discount rate. Climate change policies such as the ones over the social cost of carbon are one of the typical examples of long-term issues where benefits from policy-induced improvements can be considered as future impacts and where discounting of future costs and benefits plays a crucial role (Tol, 1999; Goulder and Stavins, 2002; Fujii and Karp, 2008; Anthoff et al., 2009).

Intertemporal preferences can be either observed empirically from the data set such as in (Lawrance, 1991; Warner and Pleeter, 2001; Dreyfus and Viscusi, 1995) or measured experimentally (Benzion et al., 1989; Chapman and Elstein, 1995; Coller and Williams, 1999; Harrison et al., 2010). Controlled economic experiments provide a natural framework to study time discounting in both laboratory and field conditions by enabling researchers to systematically vary the parameters to infer decision makers' preferences. Researchers of both economics and psychology fields have therefore devised various methods to experimentally measure intertemporal preferences described by the discounted-utility model (Samuelson, 1937). Despite different experimental methods available for eliciting time preferences there is no consensus on how to best measure discounting (Andreoni et al., 2015). Moreover, the discount rate is subjective to individual time preferences and its estimates vary greatly throughout the literature (Frederick et al., 2002; Coller and Williams, 1999).

We therefore collect 434 experimental estimates of the discount rate reported in 27 primary studies and review the literature quantitatively by a meta-analytical methods. Using a combined conditional estimator according to Stanley and Doucouliagos (2014) We explore the data for the presence of selective reporting, a phenomenon found to be widespread also in economics (Doucouliagos and Stanley, 2013). Furthermore, we focus on explaining the heterogeneity in discount rate estimates. Concentrating on aspects related to study design, methodology or subject pool characteristics we collect a set of 20 additional explanatory variables out of the primary studies and employ techniques of Bayesian model averaging (BMA, Raftery et al., 1997). By employing the BMA we address the model uncertainty inherent in meta-analysis and attempt

to establish whether these aspects systematically affect the reported estimates of the discount rate.

Our results suggest that a selective reporting is present within the relevant subsample of discounting literature. Published studies report only small estimates of the discount rate with high statistical precision, overestimating the discount rate parameter approximately twofold. Variation in estimates seem to be caused primarily by the experimental design of discounting tasks. We especially find the domain independence (Loewenstein et al., 2003), confirming the decisions made within domains of health or vacation being independent among each other, and also find that the time horizon of the task presented to subjects has an impact on resulting discount rate estimates.

Interestingly, we do not confirm several potential problems typically pointed out with respect to economic experiments. First, student subject pools tested in laboratories for discounting yield similar results as general population in large field experiments, disproving the rejection of external validity of experiments conducted on student samples. Second, real rewards do not significantly change discount rate estimates suggesting that hypothetical rewards are sufficient for collection of discount rate data.

The remainder of the paper is structured as follows. Section 2 reviews basic concepts of discounted-utility model and discusses the methods of discount rate elicitation. Section 3 describes our approach to data collection and presents the overview of our data set. Section 4 examines the selective reporting using meta-regressions. Section 5 investigates the sources of heterogeneity in the estimated discount rate parameters. Section 6 concludes. There are supplementary statistics and diagnostics of both meta regressions and BMA as well as various robustness checks to the analyses available in the appendices.

2 Estimating Discount Rate

Theory on intertemporal choice and discounting dates back to Irving Fisher's *Theory of Interest* (Fisher, 1930) and Paul Samuelson's article *Note on Measurement of Utility* where he postulated the discounted-utility model (Samuelson, 1937). His model was widely accepted with its central idea that concentrates various decisions about intertemporal choices into single parameter—the discount rate. Several modifications to the original discount functions was introduced since its origin to capture various irregularities such as hyperbolic (Ainslie, 1975; Mazur, 1984) or quasi-hyperbolic (Phelps and Pollak, 1968; Laibson, 1997) discounting functions. The central idea however remains unchanged.

This study does not provide a complete and critical review of neither the basic model nor further proposed modifications, we merely briefly mention its basic concepts that are used in vast literature related to our study. For more detailed regard we refer an interested reader to standard economics textbooks or other survey literature, especially to Andersen et al. (2014) or Frederick et al. (2002) and Cheung (2016).

Discounted utility model captures the time preferences of an individual, more specifically the preference of an individual for immediate utility over delayed utility, represented by her intertemporal utility function $U^t(c_t, ..., c_T)$ that can be described by functional form in equation 1:

$$U^{t}(c_{t},...,c_{T}) = \sum_{k=0}^{T-t} D(k) \cdot u(c_{t+k}),$$
(1)

where D(k) is the discount function and $u(c_{t+k})$ is cardinal instantaneous utility function ¹ that can be interpreted as individual's well-being in period t + k. The discount function D(k) represents the relative weight that an individual places in period t on her well-being in period t + k and encompasses parameter δ that represents individual's *pure rate of time preference*, i.e. her discount rate. This discount function can have different functional forms.

The standard exponential model, a well-known simple functional form used in majority of practical applications, is captured by following equation 2:

$$D(k) = \frac{1}{(1+\delta)^k} \qquad , k \ge 0 \tag{2}$$

where the discount rate *d* is simply $d^E(k) = \delta$. The key feature of this model is that the discount rate *d* is constant over time, i.e. the rate at which an individual discounts future well-being between today and tomorrow is identical to the rate that discounts future well-being between today and tomorrow in one month. Widely documented situation in which an individual has declining rate of time preference, not constant, is described as hyperbolic discounting. This generally means that the implicit discount rate over longer time horizons is lower than the implicit discount rate over shorter time horizons. A typical case out of the family of hyperbolic discounting functions proposed by Mazur (1984) is described by equation 3:

$$D(k) = \frac{1}{1 + \delta k} \tag{3}$$

whith hyperbolic discount rate $d^{H}(k) = (1 + \delta k)^{\frac{1}{k}} - 1$ (Andersen et al., 2014).

¹Standard assumptions on instantaneous utility function are: u(0) = 0, u'(c) > 0, $u''(c) \le 0$.

Phelps and Pollak (1968) further introduced a quasi-hyperbolic specification of the discount function for use in a social planner problem. This specification is described by equation 4:

$$D(k) = \begin{cases} 1, & \text{if} \quad k = 0\\ \frac{\beta}{(1+\delta)^k}, & \text{if} \quad k \ge 0 \end{cases}$$
(4)

where $\beta \leq 1$ and quasi-hyperbolic discount rate $d^{QH}(k) = \left(\frac{\beta}{(1+\delta)^k}\right)^{-\frac{1}{k}} - 1$. Characteristic feature of the quasi-hyperbolic specification is the discontinuity at time t = 0. This specification was furthermore applied to model individual agent behavior by Laibson (1997).

Several experimental methods are available for eliciting time preferences in both laboratory and field settings such as lotteries, choice lists, or bidding, and yet there is no consensus how to best measure discounting (Andreoni et al., 2015). Basic method for eliciting individual discount rates is however fairly simple in its essence—asking subjects questions whether they prefer some sure amount of money today (option A) or the same amount + X\$ tomorrow (option B), where X\$ is some positive amount.² By variation of X a researcher can infer bounds of an interval in which lies the subject's individual discount rate.³ Experiments therefore involve series of questions aligned in lists such as in the classical choice list design of (Coller and Williams, 1999) or (Harrison et al., 2002). Modifications to this basic method are further used to elicit preferences more precisely, such as variations in delay between options A and B, domain in which are the preferences revealed (money, health etc.), magnitude of the domain or nature of the rewards (hypothetical or real).

There are several types of elicitation methods according to Frederick et al. (2002) that are commonly used in the experimental literature: i) choice; ii) matching; iii) rating; iv) pricing. Most common type of elicitation is choice method where subjects are presented alternative options and are asked to simply choose between them. This method provides discount rate intervals pre-generated by the experimenter rather than precise estimates of the discount rate of specific individuals. Matching method, on contrary, provides an exact inference of individual's discount rate since she reveals her true indiference point by filling the blank field to *"equate two intertemporal options"* (Frederick et al., 2002, p.387). In rating tasks the subjects evaluate individual options by rating their attractiveness on predefined scale while in pricing tasks subjects specify their willingness to pay for individual options in which they either obtain or avoid par-

²We use the domain of money in dollar units for this simple illustrative example.

³The point of first switch to option B gives a measure of upper bound of her discount rate. We assume linear utility here for simplicity and discuss the relax of this assupption later.

ticular outcome. Rating and pricing tasks allow to manipulate time variable between subjects since immediate and delayed options are evaluated separately in contrast to choice and matching tasks.

Each method briefly described above has its further strong points and limitations. When subjects are asked to choose or evaluate multiple options at once in a standard choice list, the first or earlier choices inevitably influence choices that are made later. This procedural limitation—an anchoring effect—can be partially dealt with by employing titration procedures—exposing subjects to a sequence of different opposing anchors (Frederick et al., 2002). Timing of an outcome was found to have much lower effect when evaluating single option compared to a situation when two options occuring in different times are evaluated at once against each other (Loewenstein, 1987). Timing of two evaluating options is further argued to cause more general problem of additional risk or transaction cost imposed on the future option. Recent literature such as Harrison et al. (2005); Andersen et al. (2014) and others deal with this risk by employing a front end delay, shifting thus the immediate option to some nearer future and imposing some transaction costs also on the instant payoff.

Harrison et al. (2005) argues that standard choice tasks often executed through Multiple Price Lists (MPL) devised originally⁴ by Holt and Laury (2002) has three possible disadvantages: i) eliciting only interval responses; ii) allowing subjects to switch back and forth while moving down the list; and iii) MPL can be subject to framing effects. They therefore introduce an *iterative Multiple Price List* (iMPL) that allows the subjects to specify iteratively their choices through refined options within an interval chosen in the last option.

Inference of the discount rates out of the experimental task is dependent on the utility function presented in discounted-utility model 1. This function is however unobserved and therefore usually assumed to be linear, generating biased estimates for individuals with non-linear utility functions (Cheung, 2016). Recent literature of Andersen et al. (2008, 2014) measure time preferences controlling for non-linear utility by employment of *joint elicitation strategy*. Using the equivalence of utility for risk and time they use series of binary choices to infer the discount function conditional upon utility function elicited through Holt and Laury (2002) risk preference task. Further modifications of the design to measure time preferences controlling for non-linear utility includes e.g. work of Laury et al. (2012) who interact risk with time by using a lottery to be paid out with probability p_t in time t and with probability p_{t+k} in time t + k, where $p_t \leq p_{t+k}$ and p_{t+k} is varying through the choice list.⁵ Further ex-

⁴The MPL was originally devised for mesuring risk aversion, but used for measuring intertemporal preferences by Coller and Williams (1999) or Harrison et al. (2002)

⁵They assume expected utility under risk.

periments measuring time preferences under control for non-linear utility were done e.g. by Takeuchi (2011) who employs separate choices under risk and over time using matched pairs of payoffs; Andreoni and Sprenger (2012b,a); Andreoni et al. (2015) who examine risk and time preferences through individual elicitation methods—convex time budgets and double multiple price list tasks; or Attema et al. (2016) who introduces a *direct method* of measuring discounting that is not dependent on knowledge or measurement of utility.

An alternative method for infering the discount rates was devised by Chabris et al. (2008b) who infer the intertemporal preferences not only from standard choice tasks but also adopt an approach of inferring preferences using response times from these choices, i.e. how long it actually takes subjects to choose between option A and option B. They assume that *"subjects should take longest to decide when the two options are most similar in their discounted values"* and therefore argue that the inference from response times should, in principle, work (Chabris et al., 2008b, p.7). Results of Chabris et al. (2008b) suggest⁶ that choice-based and response-time-based estimates are nearly identical in their setting.

3 Data Set

We combine two sources of primary studies into our dataset—we search google scholar⁷ for experimental discounting literature and furthermore go through references of retrieved studies for additional discounting studies that do not get into top search results. We do not include all studies from each source exclusively. We apply two basic inclusion criteria. Each study included into our dataset must be of experimental nature, either lab or field, and need to report estimate of discount parameter—rate or factor—or contain information allowing for its calculation. We do not explicitly require the study to report a measure of uncertainty but we always search for it and code it since it is crucial for our analysis, i.e. for measuring publication bias. Furthermore, we include only published studies since they are peer-reviewed and therefore assumed to be of better quality. We add the last study on October 11, 2018, and terminate the data collection. Our final dataset covers 27 studies comprising 434 estimates of the discounting parameter.⁸ The oldest study was published in 1989, the newest

⁶This paper was published in form of a working paper in NBER working paper series only.

⁷Google Scholar searches full text of studies in addition to the title, abstract, and keywords; search querry: discount method experiment field "discount rate" OR "discount factor"

⁸We assembled 404 estimates of discount rate, either directly or calculated out of available data (e.g. regressions) and 30 estimates of discount factors from 1 study (Read and Read, 2004) that we recoded to rates according to corresponding discounting formulas. We use the term *discount rate* for all these

one in 2017, representing almost three decades of research in the area. Overview of reviewed studies is presented in Table 1.

Andersen et al. (2006)	Coller and Williams (1999)
Andersen et al. (2008)	Deck and Jahedi (2015a)
Andersen et al. (2014)	Deck and Jahedi (2015b)
Andreoni et al. (2015)	Dolan and Gudex (1995)
Bauer and Chytilová (2013)	Hardisty et al. (2013)
Benzion et al. (1989)	Harrison et al. (2002)
Cairns and der Pol (1997)	Harrison et al. (2010)
Cassar et al. (2017)	Duquette et al. (2012)
Chabris et al. (2008a)	McClure et al. (2007)
Chapman and Elstein (1995)	Newell and Siikamäki (2015)
Chapman (1996)	Read and Read (2004)
Chapman and Winquist (1998)	Tanaka et al. (2010)
Chapman et al. (1999)	Zauberman et al. (2009)
Chesson and Viscusi (2000)	

Table 1: Studies included into dataset

Apart from collection of estimates of our cardinal variable—the discount rate and available measures of statistical uncertainty—we code additional explanatory variables to control for sources of variation in our data sample. We control for the type of the discounting estimate, that is whether the estimate was reported as discount rate or factor. We include the length of time horizon presented to the subjects in the experimental task or questionnaire.⁹ In order to be able to classify discount rate estimates to different types of discounting we include categorical variable describing whether the reported estimate is of hyperbolic or exponential discounting.¹⁰ We further control whether the study is done in the lab or in the field; if payoffs used in the study are only hypothetical or real, i.e. paid out to the subjects; which elicitation method (choice, matching, rating) and domain (money, health, etc.) is used for eliciting of specific estimate; and whether the framing of the task is positive (giving), negative (loosing) or neutral. We also control for standard essentials of subject pool by includ-

estimate types in the following text.

⁹Some studies vary time horizon holding the payoff constant. Such studies do not separate different time horizons into different treatments and therefore do not report exact time point in which the subjects made the switch. We therefore code the maximum horizon presented in the task and include dummy variable to control for this effect.

¹⁰We can not code discounting type in 59 observations due to lack of information. We further recognize exponential mixture share (1 case), quasi-hyperbolic discounting (3 cases) and mixed general model (1 case) in the dataset. We respectively recode these due to low rate of incidence as: exponential, hyperbolic, na.

ing the dummy variables for whether it contains students or professionals; gender of subjects it includes (m/f/both); and continent and country where the subject pool was hired. Additionally we look up publication information for each study that comprises of impact factor¹¹ and the number of Google citations that we weight by number of years since first version of the study. We describe and further use these additional variables in Section 5.

Discount rates in our dataset ranges from -0.035 to 8.930, with mean value of 0.408 and standard deviation 0.659. Histogram of the variable discrate with upper bound imposed on the 99th percentile of the date is presented in Figure 1. The distribution is highly left-skewed, with median value of 0.235. Negative values of discount rate estimates are very rare, even though present, and exclusively a matter of health domain. The distribution thus have several outliers on both sides. We overcome potential influence of these outliers in our analysis by winsorization of discount rate¹² at 0.025 percentile level from each tail.



Figure 1: fig: Histogram of discount rate estimates

Notes: The figure depicts histogram of the discount rate estimates reported by individual studies. Extreme values are omitted from the graph ($\delta < 1$), but are accounted for in all regressions. Solid vertical line denotes the sample mean, dashed vertical line denotes the sample median.

 $^{^{11}\}mbox{We}$ use Discounted recursive REPEC impact factor that can be retrieved at https://ideas.repec.org/top/top.series.rdiscount.html

¹²Together with combined standard errors of discount rate described farther in this paragraph.

Standard errors¹³ of the discount rate estimates are reported only in 252 observations out of 434 in our dataset. Researchers often claim that their true effect is indeed large and robust even though they do not support such claims by providing the standard errors of their estimates. We do not exclude such studies that do not report any statistical measure of uncertainty of the estimates; we apply Bootstrap resampling technique¹⁴ to those studies rather than exclude them to approximate estimates of standard errors at least on the study-level. We then combine data of standard errors from primary studies with bootstrapped standard errors on study-level into variable *discrate_se_comb* that is used in our subsequent analysis. ¹⁵ Our results prove robust when we control for the effect of resampling. The subset of discount rates without reported uncertainty measures has the mean of 0.326 and standard deviation of 0.336. There is total of 197 observations out of 10 studies that does not provide any statistical measure of uncertainty of reported estimates in our dataset, which makes 45.39% of the standard errors being bootstrapped on study level. Summary statistics of selected variables are presented in Table 2. Study-level and country-level heterogeneity of discount rate estimates can be observed in Figures 2 and 3, respectively.

Table 2: Summary statistics of selected variables

Statistic	Ν	Mean	St. Dev.	Min	Pctl(25)	Pctl(75)	Max
discrate	434	0.408	0.659	-0.035	0.137	0.430	8.930
discrate_se	252	0.025	0.039	0.0001	0.008	0.026	0.390
discrate_se_comb	434	0.063	0.132	0.000	0.009	0.040	1.505

Notes: Following abbreviations stand for: discrate = discount rate; win = winsorized at p = 0.025; comb = combined with resampled std. errors.

¹³We recalculated other reported statistical measures of uncertainty to standard errors.

¹⁴We use *bootstrap* command in Stata 14.0, which we apply to discount rate data of each study separately, storing bootstrapped mean.

¹⁵There are two studies in our dataset that do not report any measure of statistical uncertainty and do not provide more than one observation per study. Resampling can be used for approximation of standard errors only if there are multiple observations per study, otherwise SE of resampled estiamte is zero. For Coller and Williams (1999); Deck and Jahedi (2015b) we therefore obtain zero standard error approximations and consequently can not calculate precision of the estimates as 1/*SE*. Further parts of our analysis that use weighting by precision of the estimate are therefore based on 25 studies only. This applies especially for Funnel plot of discount rate in Figure 4 and Funnel asymmetry tests in Table 3.



Figure 2: Study-level variation of discount rate estimates

Notes: The figure shows a box plot of estimates of discount rate reported in individual primary studies (observations above the 99th percentile omitted). Red solid line represents sample mean.

4 Detection of Publication Bias

Selective reporting has been identified as serious threat to scientific validity.¹⁶ Researchers thrive for scientific publications and therefore tend to have strong preference for reporting statistically significant or "positive" results. Selective reporting was found to be widespread also in economics (Doucouliagos and Stanley, 2013). We hypothesize that discounting literature is no exception. Selective reporting is a general term for publication selection or bias, a special case that limit its essence to published studies only. We are more prone to the term publication bias since we include only

¹⁶See for example Stanley and Doucouliagos (2014); Havranek et al. (2018b) and others for lists of references that address to selective reporting.

Figure 3: fig: Country-level variation of discount rate estimates



Notes: The figure shows a box plot of estimates of discount rate reported in primary studies executed in individual countries (observations above the 99th percentile omitted).. Red solid line represents sample mean.

published studies into our dataset and therefore use this term thenceforth.

Discount rate express intertemporal preferences of an individual—the lower the discount rate, the more patient one is and vice versa. General belief among economists and psychologists is that the discount rate estimates should be positive¹⁷ between zero and one, with a tendency to impatience. Estimates outside this range together with statistically insignificant estimates are therefore reported rarely in the literature. Our sample mean of 0.408 indeed implies such expectations. One would further expect that there would be certain portion of estimates reported with appropriate statistical precision in an empirical literature, which the discounting one certainly is. Figure 4 however shows different pattern. It is a so-called funnel plot that commonly serves as a visual representation of publication bias (Egger et al., 1997; Stanley and Doucouliagos, 2010). This simple scatter plot with the effect's magnitude on the horizontal axis and the inverse of the standard error-its precision-on the vertical axis, shows that researchers report precise data of discount rate almost exclusively for small effect sizes only and that high effect sizes of discount rate are associated with low statistical precision. This fact certainly suggest presence of publication bias in our sample of discounting data (Stanley, 2005; Stanley and Doucouliagos, 2014).

Next, we examine correlation between the discount rate estimates and their standard errors quantitatively to test for presence of publication bias. Following Stan-

¹⁷Sign of discount rate estimate depends highly on the framing of experimental task—gains having generally positive sign while losses related specifically to health domain tend to report negative sign of the estimate. Negative estimates are, however, rare in our sample, the funnel plot is asymmetric denoting only positive side of the graph since the majority of experimental tasks are offering monetary gains (77, 19%).

Figure 4: Funnel of discrate



Notes: The figure depicts funnel plot of the discount rate estimates. Extreme values are omitted from the graph ($\delta < 1$), but are accounted for in all regressions. Solid vertical line denotes the sample mean. Based only on 25 studies due to inability of calculation of precision for Coller and Williams (1999); Deck and Jahedi (2015b), more information in footnote 15.

ley and Doucouliagos (2014) we apply a combined conditional estimator *PET-PEESE*. First, we test H_0 : $\hat{\delta}_1 = 0$ by a simple meta-regression model used by Egger et al. (1997)—the *Egger* or *precision-effect* test in equation 5:

$$\hat{\delta}_{ij} = \delta_1 + \gamma_1 \cdot SE(\hat{\delta}_{ij}) + u_{ij},\tag{5}$$

where $\hat{\delta}_{ij}$ is the *i*-th estimate of the discount rate and $SE(\hat{\delta}_{ij})$ is its corresponding standard error of the *j*-th study; u_{ij} is the disturbance term. The precision-effect test (PET) "provides a valid basis for determining whether there is a genuine empirical effect beyond publication selection bias" (Stanley and Doucouliagos, 2014, p.64) and is known to give biased estimates of the underlying true effect when this is non-zero (Stanley, 2008). First part of Table 3 shows results of PET test for various model specifications; we cluster standard errors at the study level in all of them. We obtain positive and statistically significant estimate of γ_1 —the coefficient that represents publication bias—in all specifications. The underlying discount rate corrected for the bias represented by the coefficient δ_1 is 0.258 in OLS specification, which is about 62% of original mean of our sample (0.408). Such result is consistent with the rule of thumb suggested by Ioannidis et al. (2017) and indicates strong publication bias. We do not obtain statistically significant underlying effect when controlling for study-specific characteristics by study-level fixed effects.

PET tests	OLS	FE	Precision	Study
SE (publication bias)	1.923**	5.046***	4.958**	1.661***
	(0.569)	(0.836)	(1.718)	(0.491)
Constant (corrected effect)	0.258***	0.0790	0.0796*	0.313***
	(0.0615)	(0.0479)	(0.0352)	(0.0703)
Observations	434	434	432	434
PEESE tests	OLS	FE	Precision	Study
SE^2 (publication bias)	5.173**	15.17*	10.36**	4.821**
	(1.845)	(6.189)	(3.487)	(1.564)
Constant (corrected effect)	0.310***	0.198**	0.114**	0.350***
	(0.0724)	(0.0694)	(0.0415)	(0.0672)
Observations	434	434	432	434

Table 3: Funnel asymmetry tests

Notes: The table reports results of the regressions 5 (PET test) and 6 (PEESE test). All variables winsorized at p = 0.025. The standard errors of the regression parameters are clustered at the study level and shown in parentheses. OLS = ordinary least squares, FE = study-level fixed effects. Precision = weighted by the inverse of the standard error. Study = weighted by the inverse of the number of estimates reported per study. Weighting by the precision of the study based only on 25 studies due to inability of calculation of precision for Coller and Williams (1999); Deck and Jahedi (2015b), more information in footnote 15. * p < 0.10, ** p < 0.5, *** p < 0.01.

Since the equation 5 contains heteroskedasticity,¹⁸ we further use weights in order to estimate the meta-regression model in line with Stanley and Doucouliagos (2014). First, we use *precision* of the estimate, i.e. 1/SE, as weights.This approach gives more weight to more precise results, but there is an important caveat in this weighting scheme. Estimation of standard errors is in economics often an important feature of the model and weighting by precision can create bias by itself in case the study underestimates the standard error. We therefore use also the inverse of the number of estimates reported per study as second weight (*study*). This approach equalizes the impact that each study has on the results. Both weighting schemes yield positive and significant results for both publication bias as well as corrected effect. Weighting by

¹⁸There is apparent heteroskedasticity in the baseline regression, since the standard error of the discount rate estimate, i.e. the independent variable, is a measure of the dispersion of the magnitude of the estimate of the discount rate, i.e. dependent variable.

the inverse per study estimates gives milder effect on the corrected discount rate resulting in about 76% of the original mean of our sample.

Regardless of insignificances described above, Table 3 implies rejection of the PET hypothesis and indicate that the true effect in the equation 5 is indeed non-zero. Since the true model of the selective reporting is an undisclosed complex function of the standard errors, Stanley and Doucouliagos (2014) examines a quadratic approximation that yields very good results. We therefore apply a second step in line with Stanley and Doucouliagos (2014) in our analysis of discounting literature—a conditional, weighted quadratic estimator of precision-effect estimate with standard errors (PEESE), described in equation 6:

$$\hat{\delta}_{ij} = \delta_2 + \gamma_2 \cdot SE^2(\hat{\delta}_{ij}) + u_{ij},\tag{6}$$

where δ_1 from equation 5 is constrained to zero.

Second part of the Table 3 reports results of equation 6 in different specifications. We again cluster standard errors of all regressions on the study-level since estimates reported in the same study are unlikely to by independent. All model specifications yield positive and significant results of publication bias as well as underlying corrected effect. Numerically are our results consistent with the true effect of the discount rate between 0.11 and 0.35 compared to the original mean of the data being 0.408. Relevant discounting literature therefore overestimates the discount rate parameter due to the publication bias approximately twofold.¹⁹ This degree suggests a strong publication bias according to the rule of thumb presented by Ioannidis et al. (2017). Our results prove to be robust when controlling for resampling of missing standard errors. Table 6 in the appendix provides results of meta-regressions performed on observations with reported uncertainty measures.

5 Heterogeneity in Estimates

Heterogeneity in estimates of the discount rate was described by previous studies (Frederick et al., 2002; Andersen et al., 2014; Cheung, 2016). Frederick et al. (2002, p.352) claims that: *"While the DU model assumes that people are characterized by a single discount rate, this literature reveals spectacular variation across (and even within) studies."* Our dataset is no exception—Figures 2 and 3 show that we observe strong heterogeneity on both study-level as well as country-level. In this section we try to tackle the problem and explain high variation in estimates by analysing discount rates and their

¹⁹Average true effect resulting from different specifications is $0.11 + \frac{0.35 - 0.11}{2} = 0.23$, being approximately one half of the reported mean estimate of the discount rate.

standard errors together with additional 20 explanatory variables we have coded out of the examined studies.

Common approach would be to estimate model with all explanatory variables by some standard regression technique (OLS) and eliminate potential redundant variables in a step-wise fashion. This method would however not prove its usefulness in our case since standard errors of many coefficients in such regression would be exaggerated and therefore would not provide good-enough basis for elimination. Furthermore, even though there are some theoretical predictions stemming from the discounting literature regarding the decisions about variables that could be included in the model, we gathered a lot of information that would be hard to classify. It would be hard to identify a proper set of variables, since we face extensive model uncertainty. We therefore search for some technique that can address this uncertainty inherent in a meta-analysis—we estimate the heterogeneity of our dataset by Bayesian model averaging (Raftery et al., 1997).

Bayesian model averaging (BMA) tackles the problem of uncertainty by estimating models for all possible combinations of explanatory variables in the dataset²⁰ and constructing a weighted average over estimated coefficients (posterior means) across all these models. Weights used for averaging stem from posterior model probabilities arising from Bayes' theorem and are analogous to information criteria in frequentist econometrics. Posterior model probabilities (PMP) measure how well the particular model fits the data conditional on model size. BMA produces posterior inclusion probability (PIP) for each variable, which is the sum of the posterior model probabilities for the models in which the variable is included. Recent application of Bayesian model averaging in meta-analyses in economics and finance comprise of e.g. Havranek et al. (2015b, 2017); Havranek and Irsova (2017); Havranek et al. (2018a,b). More details on BMA can be found e.g. in Raftery et al. (1997) or Eicher et al. (2011).

Application of BMA is however not straightforward since estimating all possible combinations of models with standard available technology becomes infeasible starting from already a few number of variables. Typical approach is to approximate the whole model space by applying the Markov chain Monte Carlo Model Composition algorithm that walks only through the models with high posterior model probabilities (Madigan et al., 1995). This approximation can be done relatively simply by using the BMA **R** package developed by Zeugner and Feldkircher (2015). We take advantage of this package and estimate the model space using $2 \cdot 10^6$ iterations and $1 \cdot 10^6$

²⁰If matrix of explanatory variables *X* contains *K* potential variables, this means estimating 2^{K} variable combinations, i.e. 2^{K} models. This result in $2^{21} = 2097152$ models in our case, since we coded the discount rate, its standard errors (1) and 20 additional variables.

burn-ins to ensure convergence.²¹ Bayesian model averaging is sensitive to estimation framework, especially to the use of *priors* representing researcher's prior beliefs on probability of each model (the model prior) and regression coefficients (Zellner's g-prior). In the baseline specification we decide to follow Eicher et al. (2011) who show that these intuitive priors yield the best predictive performance: we use the unit information prior ("UIP") for Zellner's g-prior that assigns the prior weight of one data observation; and the uniform model prior that gives each model the same prior probability.²²

5.1 Variables Explaining Heterogeneity

We have coded 20 additional explanatory variables that should help us to explain the heterogeneity in discount rate estimates. The explanatory variables are listed in Table 4; we report description of each variable, its mean, standard deviation, and mean weighted by the inverse of the number of estimates reported per study, which effectively equalizes the impact of small and large study samples. We divide the explanatory variables into 4 categories: Estimation characteristics, Experimental characteristics, Subject pool characteristics, and Publication characteristics.

5.1.1 Estimation Characteristics

Variation of discount rate estimates can in the first instance stem from theoretical assumptions of intertemporal choice model used in the experimental task presented to subjects. That is mainly from type of discounting model and time horizon that subjects face in their decision. We provide survey of discounting models in Section 2. Research studies included in our dataset report hyperbolic discounting model most frequently (281; 65%), followed by exponential discounting model (89; 20%). Special cases of discounting models such as exponential mixture share, quasi-hyperbolic discounting or mixed general model occur rarely in our dataset (5 cases in total). We could not identify discounting model in non-negligible 14% of the data (59) and use it as the reference category. Time horizon of the decisions presented to the subjects spans from one day to 50 years, while its mean value is 5.01 years. We therefore apply logarithm value to this variable in order to dispose of scale effect. Some studies, however, vary time hori-

²¹Regarding iterations (*I*) and burn-ins (*B*) of BMA we compared three scenarios: i) $I = 10 \cdot 10^5$, $B = 5 \cdot 10^5$; ii) settings indicated above and finally used in the analysis; and iii) $I = 10 \cdot 10^6$, $B = 5 \cdot 10^6$. All scenarios yielded very similar results.

 $^{^{22}}$ Robustness check using benchmark g-prior for parameters suggested by Fernández et al. (2001) and the beta-binomial model prior for the model space according to Ley and Steel (2009) can be found in the appendix C

Variable	Description	Mean	SD	WM
discrate discrate_se_comb	Discount rate estimate. Standard error of the discount rate estimates combined with resampled standard errors.	0.407 0.063	0.659 0.132	0.072 2.789
Estimation characteris	tics			
disc_factor	= 1 if reported variable is discount factor in- stead of rate.	0.069	0.254	0.033
hyperbolic	= 1 if discounting type estimated is hyperbolic.	0.654	0.476	0.046
exponential	= 1 if discounting type estimated is exponential.	0.207	0.406	0.071
In_timelength	Logarithm of time horizon of the task.	0.721	1.529	-0.022
idtimehorizon	= 1 if only maximum time horizon of the task could be coded for observation.	0.357	0.480	0.083
method_exp	= 1 if controlled laboratory experiment was used instead of a field experiment.	0.764	0.425	0.056
Experimental characte	ristics			
reward_real	= 1 if the reward subjects received was real.	0.384	0.487	0.109
elicmethod_match	= 1 if matching was used as a method of elicita- tion.	0.495	0.501	0.029
domain_health	= 1 if experiment was conducted over the do- main of health	0.105	0.308	0.062
domain_other	= 1 if experiment was conducted over other do- main (vacation, liquid or certificate)	0.096	0.296	0.057
framing_negative	= 1 if framing of the experimental task was pre-	0.041	0.200	0.122
framing_neutral	= 1 if framing of the experimental task was pre- sented as neutral.	0.027	0.164	0.062
Subject pool character	istics			
ln_samplesize	Logarithm of the sample size used for the experiment.	4.796	0.752	0.062
students	= 1 if subject pool consisted of students.	0.375	0.485	0.069
male	= 1 if subject pool contained male subjects only.	0.025	0.157	0.051
female	= 1 if subject pool contained female subjects	0.029	0.171	0.061
	only.			
continent_namerica	= 1 if experiment was conducted in North America.	0.306	0.462	0.112
continent_asia	= 1 if experiment was conducted in Asia.	0.110	0.314	0.062
Publication characteris	stics			
ln_google_per_year	Logarithm of the number of citations the study	2.739	1.340	0.058
std_pubyear	Standardized publication year of the study.	0.000	1.001	NA

Table 4: Summary of explanatory variables

Notes: SD = standard deviation, WM = mean weighted by the inverse of the number of estimates reported per study.

zon holding the payoff constant and therefore do not separate different time horizons into different treatments. Since such studies do not report exact time point in which the subjects made the switch, we cannot always distinguish the exact time horizon corresponding to the estimate. We therefore code at least the maximum time horizon presented to subjects in the task and include dummy variable to control for the effect of this inconsistency in the data. Last but not least we control for an experimental method of estimation—i.e. if controlled laboratory experiment or a field experiment was used in the certain study.

5.1.2 Experimental Characteristics

Any experiment can be affected by procedural subtleties. Second set of explanatory variables therefore comprise of experimental and behavioral characteristics of the task presented to the subject pool. Psychological research suggest that there should be no systematic difference observed between real and hypothetical payoffs in discounting experiments (Johnson and Bickel, 2002; Kühberger et al., 2002; Locey et al., 2011). Most recent literature provide more ambivalent results stating that hypothetical conditions yield patterns of discounting that mirror those for real effort tasks but these may change with repeated exposure to the decisions-nature of the payoffs provided together with repetition of those tasks therefore need to be taken into account when designing discounting studies (Hinvest and Anderson, 2010; Malesza, 2018). We therefore control for this payoff effect by extracting the information on the nature of the reward from primary studies. 62% of our observations is reported with only hypothetical payoffs. This information can furthermore serve as a proxy for the size of the payoffs presented to subjects, since large payoffs are often associated with hypothetical decisions and comparatively small payoffs with real decisions (Kühberger et al., $2002).^{23}$

Following Frederick et al. (2002) and others we control for the variation in estimates caused by the elicitation method used in the experiment. We include dummy variables for matching tasks into our set of explanatory variables, taking choice tasks as reference category present in 49% of cases. Important behavioral aspect of the task is represented by the domain over which is the intertemporal decision made. The majority of observations utilize monetary payoffs (82%); we therefore take is as natural reference category in this regard. We code the remaining domains in dummy variables, distinguishing between health domain (10.5%) and others (vacation, liquid or

²³We could not code the magnitude of the payoffs out of the primary studies and therefore use only proxy instead.

certificate—9.6%). ²⁴

The design of any experiment is seldom immune to issues of framing effects that refer to the finding that subjects often respond differently to different descriptions of the same problem (Tversky and Kahneman, 1981). The majority of discounting tasks are presented (framed) as positive monetary decision, i.e. choice between some sure amount of money today or some greater amount tomorrow (93%). There are however also negative framing of the tasks present in our dataset (4%), where Chapman and Winquist (1998) and Hardisty et al. (2013) used monetary losses in their experiments. Other studies with negative framing operate with health domain (Dolan and Gudex, 1995; Read and Read, 2004). We coded neutral framing for only 2% of observations.²⁵

5.1.3 Subject Pool Characteristics

We describe subject pool characteristics of an individual study by several variables. First, we control for the size of the subject pool by coding the number of subjects used for deriving the respective observation. We use logarithmic scale to reduce its size effect on estimation, true mean value is equal to 160.7. Second, we control for composition of subject pool by incorporating dummy variables stating whether it consists of male and female subjects. The majority of studies however uses non-exclusive subject pools consisting of both males and females (94.5%). General concern of any experimental study is its external validity, i.e. the extent to which its results can be generalized to other situations. Economic experiments are often criticized for using university students of (mainly) economics as experimental subjects-a pool of specific characteristics not always generalizable to the whole population (Marwell and Ames, 1981; Carter and Irons, 1991; Frank et al., 1993). Behavior of decision makers recruited from natural markets has been examined in variety of contexts and it has, on the other hand, typically not differed from that exhibited by more standard (and far less costly) student subject pools (Davis and Holt, 1993, p.17).²⁶ We therefore control for the effect of student subject pool in our analysis. Last but not least we hypothesize that the heterogeneity in estimates may be caused by different socio-economic areas and therefore

²⁴Total of domain percentages do not sum to 100% since some experiments used more domains at once. Money was used together with health in 8 observations (Chapman and Elstein, 1995; Chapman, 1996) and triad money, health and vacation was used in 3 observations (Chapman, 1996).

²⁵Neutral framing occur only for Chapman (1996). In this study participants made decisions that traded money for health and vice versa in both positive and negative framing. Since we could not distinguish individual treatments within reported results, we coded framing as neutral.

²⁶See (Davis and Holt, 1993) for examples of this evidence. More recent literature on non-significant differences between student and non-student samples represent e.g. (Depositario et al., 2009).

control for continents out of which the subject pool was recruited.²⁷

5.1.4 Publication Characteristics

Furthermore, we include several publication characteristics, such as in Havranek et al. (2018a) or Havranek et al. (2018b), in order to control for systemic variation of discount rate estimates with respect to year of publication or quality of the study. Controlling whether there is a systemic difference between old and new studies can be easily done by including the year of publication into the analysis. We include only published studies into our dataset and use standardize their publication year.²⁸ Addressing quality of the study is of higher difficulty. We derive information about the quality of the study from Discounted recursive REPEC impact factor²⁹ which is not available for studies outside the field of economics. We therefore gather the number of Google Scholar citations for each study and the year of first version of the study reported by Google Scholar. Then we construct the proxy for quality by weighting the number of citations by number of years since first version of the study. We again use logarithmic scale to reduce its size effect on estimation.

5.2 Results of BMA

The results of the BMA estimation are visualized in Figure 5. The variables are displayed on the vertical axis sorted by posterior inclusion probability from top to bottom. PIP can be thought of as a Bayesian analogy of statistical significance—we therefore see the most significant variables on top of the figure. Columns on the horizontal axis denote individual regression models sorted cumulatively according to the posterior model probability (PMP) from left to right. The PMP represents how well the model fits the data relative to its size; the width of the columns is proportional to the PMP. The colors of individual cells represent signs of corresponding regression coefficients. Blue color (darker in greyscale) depicts included variables with positive estimated sign while the red color (lighter in greyscale) depicts included variables with negative estimated sign. Variables with no colors are not included in given model.

The numerical results of BMA are reported in the left-hand panel of Table 5, which

²⁷The majority of studies however recruit subjects from European countries (58.3% obs.), and North American area (30.6% obs.). Only 11.1% observations originate from Asian continent. Ethnic and cultural diversity is therefore not large.

²⁸We standardize by subtracting mean and dividing by standard deviation. We originally used logarithm of the publication year but this specification caused severe deflections of intercepts since due to strongly positive nature of publication year. We therefore use standardization instead.

²⁹Discounted recursive REPEC impact factor can be retrieved at https://ideas.repec.org/top/top.series.rdiscount.html



Figure 5: Model inclusion in BMA (based on "UIP" g-prior)

Notes: Notes: The response variable is the estimate of discount rate. Columns denote individual models; variables are sorted by posterior inclusion probability in descending order. The horizontal axis denotes cumulative posterior model probabilities; only the 5.000 best models are depicted. Estimation based on "UIP" Zellner's *g-prior* and "uniform" model probability prior recommended by Eicher et al. (2011). Blue color (darker in greyscale) depicts included variables with positive estimated sign. Red color (lighter in greyscale) depicts included variables with negative estimated sign. Variable with no color is not included in the model. Numerical results of BMA exercise are reported in Table 5.

reports posterior mean and standard deviation for each variable together with posterior inclusion probability. We follow the rule of thumb presented in Kass and Raftery (1995), according which there is a decisive evidence that three of our regressors have an effect on the discount rate estimate with PIP higher than 0.99.³⁰ The most important variables explaining heterogeneity in discount rate estimates seem to be the domain over which subjects of the experiments are making particular decisions and the length of the delay or time horizon presented in those decisions.

Values of all future outcomes should be discounted at a constant rate according to the discounted utility theory (Frederick et al., 2002). Constant rate of discounting is violated e.g. by hyperbolic discounting (Mazur, 1984) or more recently Tsukayama and Duckworth (2010) found that subjects discounted rewards more steeply when they found the domain particularly tempting. The second part of this statement, the equivalence of all future outcomes, refer particularly to the domain over which is discounted. Several studies found that discount rates are not well correlated across different domains such as money and health, which is called domain independence. The signs of the effects however differ. Cairns (1992) for example estimated discount rates significantly lower for future health as compared to future wealth states; Chapman and Elstein (1995) on the other hand demonstrated on two experiments that decision makers use different discount rates for health-related decisions and money-related decisions, but with higher rates for health domain.³¹ Our results confirm domain independence. Both health and "other" domains, specified in our dataset as vacation, special certificates or even liquids, have positive effect on individual discount rate estimate suggesting that individuals tend to be more impatient when making decisions over their health states or holiday preferences then over money.

Second important factor influencing the heterogeneity in discount rate estimates is the length of the delay over which is the decision made. This factor is inherently embedded as the parameter k in the discounted utility model presented in equation 1. We again repeat that the discounted utility theory states that values of all future outcomes should be discounted at a constant rate (Frederick et al., 2002) and that this property is violated by several anomalies such as hyperbolic discounting. Our results confirm that the discount is indeed not constant. There is a small negative effect of the length of the delay on discount rate apparent in our dataset, therefore the longer the time horizon the lower the estimated discount rate. We hypothesized that this result would be influenced by our inability to code exact time horizon with which the par-

³⁰Furthermore, one variable has the evidence strong (*PIP* \in (0.95; 0.99)) and two variables week (*PIP* \in (0.5; 0.75)).

³¹Follow e.g. Loewenstein et al. (2003) for more examples of domain independence.

	Bayesian Model Averaging			Freque	entist Cheo	ck (OLS)
Variable	Post. Mean	Post. SD	PIP	Coeff.	SE	p-val.
Intercept	0.192	NA	1.000	0.263	0.045	0.000
Standard error	1.446	0.230	1.000	1.254	0.096	0.000
Estimation characteristic	CS					
disc_factor	-0.218	0.165	0.716			
hyperbolic	0.003	0.022	0.063			
exponential	-0.005	0.029	0.068			
ln_timelength	-0.074	0.028	0.957	-0.082	0.022	0.000
idtimehorizon	0.013	0.049	0.117			
method_exp	0.036	0.095	0.182			
Experimental characteris	stics					
reward_real	0.005	0.040	0.072			
elicmethod_match	0.036	0.124	0.142			
domain_health	0.369	0.100	0.993	0.183	0.166	0.270
domain_other	0.772	0.105	1.000	0.533	0.205	0.009
framing_negative	-0.001	0.032	0.045			
framing_neutral	0.003	0.040	0.046			
Subject pool characterist	ics					
ln_samplesize	0.000	0.009	0.047			
students	-0.000	0.021	0.055			
male	-0.002	0.039	0.046			
female	-0.006	0.046	0.057			
continent_namerica	-0.021	0.079	0.114			
continent_asia	0.130	0.151	0.501	-0.061	0.188	0.745
Publication characteristi	CS					
ln_google_per_year	0.005	0.016	0.131			
std_pubyear	0.023	0.070	0.172			

Table 5: Explaining heterogeneity in discount rate estimates

Notes: The frequentist check (OLS) includes variables recognized by the (BMA) as comprising the best model. Standard errors are clustered at the study level. We do not include variable disc_factor into OLS estimation since discount factor was reported only in one study and we would therefore not be able to cluster the estimation on the study level. SD = standard deviation, PIP = Posterior inclusion probability, SE = standard error.

ticular discount rate estimate was observed for all observations³². This however does not seem to prove as right expectation. The dummy variable identifying that the study reports only maximum time horizon of the task is not included in many BMA models. There is therefore not a significant difference between observations from different time ranges of the discounting tasks, or more specifically there is not a significant difference between experimental treatments varying payoffs and time horizon.

Further results suggest that it matters whether the researcher implies the discounting parameter as rate or factor. Even though we do not consider this result as robust since we examined only one study that uses factors (Read and Read, 2004).³³ A weak significance is also obtained within a dummy variable for estimates obtained on Asian continent. The benchmark demographic area—Europe—together with North America seems to report very similar pattern of discounting. Even though not considerably, Asian perception of intertemporal preferences seem to differ according to our analysis. Wang et al. (2016, p. 9) refers to the typically high long term orientation score in East Asian countries, especially Confucian cultures, who tend to put higher value on the future and are more likely to be patient.

Last but not least a measure of statistical uncertainty of the estimate—the standard error—has significant impact on the estimate of discount rate. We are well aware that our combined standard errors are artificially constructed to quite large extent.³⁴ We therefore provide a robustness check by estimating the BMA on dataset with noncombined standard errors, i.e. only with observations coded out of studies that provide a measure of statistical precision by itself, without the necessity of resampling. Results of this partial analysis that can be found in the appendix C are slightly different to those obtained by our original dataset but generally confirms the baseline.

Analysis of regressors without having an effect on discount rate estimates (i.e. with posterior inclusion probability lower than 0.5, (Kass and Raftery, 1995)) brings additional valuable insights. In Section 5.1.1 we referred to the literature suggesting there should be no difference whether real or hypothetical payoffs are used in discounting experiments.³⁵ Our results confirm that it indeed does not matter whether the decision is made in the laboratory with fictive payoffs only. Real rewards does not significantly change estimates of discount rate. Researchers can thus use hypothetical questions and rewards that have even some advantages in the domain of time preferences since they allow to ask questions involving a long time horizons and large payoffs (Wang

³²We could not code exact horizon in 35% of observations. Detail reported in Table 4.

³³We also did not include the variable disc_factor into the frequentist check regression described in the following text, for details see below.

 $^{^{34}}$ For 45.39% of observations, for more details see Table 2.

³⁵For further supporting literature see e.g. Wang et al. (2016).

et al., 2016). Neither negative nor neutral framing of decisions presented to subjects does result with high posterior inclusion probabilities. We hypothesize that this fact is caused by insufficient number of observations with these types of framing yielding low statistical power.³⁶

Our data suggest that subject pools consisting of students yield in discounting tasks similar results as general population. The rejection of external validity of studies conducted on student samples therefore does not apply in our case. Neither solely male nor female pools report significantly different results of discount rates in our sample.³⁷

5.2.1 Robustness Checks

In order to check whether our results are robust we perform several different reestimations. First, we combine the reduction of model uncertainty from BMA estimation and traditional frequentist econometrics. Best BMA model includes six explanatory variables (plus intercept) with posterior inclusion probability higher than 0.5 that should according to Kass and Raftery (1995) have non-negligible impact on our response variable. We re-estimate this best BMA model with standard OLS technique, clustering standard errors at the study level and weighting the regression by inverse number of estimates per study to reduce the impact of large studies. Results of this estimation are provided in the right hand panel of Table 5, labelled as Frequentist Check. Eventually we did not include the variable disc_factor into estimation. The estimated covariance matrix of moment conditions was not of full rank within this model since discount factor dummy variable was equal to one just and only for one study (Read and Read, 2004) and therefore caused problems within clustering at the study level.

Next, we perform a robustness check using an alternative set of BMA priors, employing the benchmark parameter g-prior suggested by Fernández et al. (2001) together with the beta-binomial model prior for the model space, which gives each model size equal prior probability (Ley and Steel, 2009).³⁸ The results of this robustness check are reported in Table 8 in the appendix, being very similar to the baseline specification.

Finally, we also perform a robustness check by estimating the BMA on dataset with non-combined standard errors already described in section 5.2. Its results are quite similar to the baseline specification but includes slightly different set of explanatory

³⁶Negative framing - 4%; neutral framing - 2%.

³⁷Even though we may not have enough data for obtaining a strong estimates with observations only slightly below 3% of the dataset.

³⁸We label this estimation according to the g-prior parameter as "BRIC."

variables. Main difference lies in inclusion of exponential model specification together with average number of Google citations obtained per year as significant in explaining heterogeneity in discount rate estimates. Interestingly, health domain disappears from variables with important impact on discount rate. Numerical as well as graphical results can be found in the appendix C.

6 Concluding Remarks

We quantitatively survey a sample of literature experimentally estimating discount rates of individual economic agents. We examine 434 estimates of the discount rate reported in 27 primary studies. By meta-regression methods commonly used in economics, medicine and other fields we detect presence of selective reporting in this literature. Mean value of reported discount rate estimates is 0.408 in our sample. We find that this value drops to about one half when we correct for publication bias, suggesting that individuals are on average more patient than indicated by examined primary studies.

Estimates of the discount rate are furthermore very disperse. We try to explain this heterogeneity by employing Bayesian model averaging, a method accounting for model uncertainty inherent in meta-analysis caused by high number of explanatory variables. We find that a crucial factor influencing the variation in discount rate estimates is the domain over which individuals make the decisions and confirm thus domain independence found by previous literature on discounting. The heterogeneity is further caused by the time horizon the individual subjects face during the decision.

The results of our study may apply in various contexts. One of the crucial dimensions over which the utility functions are defined is the time preference. Understanding individual time preferences and their synthesis to more general level is therefore a critical element of much of economic analyses. Discount rate, the parameter expressing time preferences, has an important implications in decisions regarding savings, educational investments, smoking, exercise and other real-world behaviors (e.g. Chabris et al., 2008a; Meier and Sprenger, 2010). Accurate measures of discounting parameters can provide helpful guidance in welfare analyses on the potential impacts of policies and provide useful diagnostics for effective policy targeting (Andreoni et al., 2015) or can be furthermore applicable to model political campaigns, advertisement or R&D Investments (Deck and Jahedi, 2015b). Interesting examples provide behavioural applications examined e.g. by Deck and Jahedi (2015a) and Deck and Jahedi (2015b) who examine discounting in strategic settings such as auctions or experimental contests in which it is often critical to accurately predict behavior of counterparts to act optimally. Such real-time decisions involve correct appraisal of the discount rate that can be provided by our results.

The climate change policies in which an individual pure rate of time preference or social discount rate is needed for evaluation of long term effects of those policies can serve as an example of welfare analysis application. Pure rate of time preference together with growth rate of per capita consumption and elasticity of marginal utility of consumption creates basis for calculation of Ramsay discount rate consisting of *time* and *growth* discounting elements Fearnside (2002); Anthoff et al. (2009); Foley et al. (2013). Our discount rate synthesis together with results of Havranek et al. (2015a) who synthesise the elasticity of marginal utility of consumption can be utilized for calculation of the pure rate of time preference out of the Ramsay discount rate.

We point out three possible caveats that may apply to results of our analysis. First, our results are based only on sub-sample of all available studies eliciting intertemporal preferences of individuals. The economic and psychological literature on discounting is vast. We believe that we have collected representative sample of experimental literature published with time span of almost thirty years of research. Next, only about a half of collected estimates are reported with a measure of uncertainty from which we can compute approximate standard errors. We overcome this caveat by resampling standard errors on study level for observations with missing data. Last but not least, even though we control for variation in major design aspects of experiments used for elicitation of discount rate estimates, those experiments have many unique methodological as well as procedural details that makes them different from each other. Such details are however not collectable out of primary studies and therefore unobserved by our analysis.

7 Bibliography

- Ainslie, G. (1975). Specious reward: a behavioral theory of impulsiveness and impulse control. *Psychological bulletin*, 82(4):463–496.
- Andersen, S., Harrison, G. W., Lau, M. I., and Rutström, E. E. (2006). Elicitation using multiple price list formats. *Experimental Economics*, 9(4):383–405.
- Andersen, S., Harrison, G. W., Lau, M. I., and Rutström, E. E. (2008). Eliciting risk and time preferences. *Econometrica*, 76(3):583–618.
- Andersen, S., Harrison, G. W., Lau, M. I., and Rutström, E. E. (2014). Discounting behavior: A reconsideration. *European Economic Review*, 71:15–33.
- Andreoni, J., Kuhn, M. A., and Sprenger,
 C. (2015). Measuring time preferences:
 A comparison of experimental methods. *Journal of Economic Behavior & Organization*, 116:451–464.
- Andreoni, J. and Sprenger, C. (2012a). Estimating Time Preferences from Convex Budgets. *American Economic Review*, 102(7):3333–3356.
- Andreoni, J. and Sprenger, C. (2012b). Risk Preferences Are Not Time Preferences. *American Economic Review*, 102(7):3357–3376.
- Anthoff, D., Tol, R. S. J., and Yohe, G. W. (2009). Risk aversion, time preference,

and the social cost of carbon. *Environmental Research Letters*, 4(2):024002.

- Attema, A. E., Bleichrodt, H., Gao, Y., Huang, Z., and Wakker, P. P. (2016). Measuring discounting without measuring utility. *American Economic Review*, 106(6):1476–1494.
- Bauer, M. and Chytilová, J. (2013). Women, Children and Patience: Experimental Evidence from I ndian Villages. *Review of Development Economics*, 17(4):662–675.
- Benzion, U., Rapoport, A., and Yagil, J. (1989). Discount rates inferred from decisions: An experimental study. *Management science*, 35(3):270–284.
- Cairns, J. and der Pol, M. (1997). Constant and decreasing timing aversion for saving lives. *Social Science & Medicine*, 45(11):1653–1659.
- Cairns, J. A. (1992). Health, wealth and time preference. *Project Appraisal*, 7(1):31–40.
- Carter, J. R. and Irons, M. D. (1991). Are Economists Different, and If So, Why? *Journal of Economic Perspectives*, 5(2):171–177.
- Cassar, A., Healy, A., and Von Kessler, C. (2017). Trust, risk, and time preferences after a natural disaster: experimental evidence from Thailand. *World Devel*opment, 94:90–105.

- Chabris, C. F., Laibson, D., Morris,
 C. L., Schuldt, J. P., and Taubinsky,
 D. (2008a). Individual laboratorymeasured discount rates predict field behavior. *Journal of risk and uncertainty*, 37(2-3):237.
- Chabris, C. F., Laibson, D. I., Morris, C., Schuldt, J. P., and Taubinsky, D. (2008b). Measuring intertemporal preferences using response times. *NBER Working Papers*.
- Chapman, G. B. (1996). Temporal discounting and utility for health and money. *Journal of Experimental Psychol*ogy: Learning, Memory, and Cognition, 22(3):771.
- Chapman, G. B. and Elstein, A. S. (1995). Valuing the future: Temporal discounting of health and money. *Medical decision making*, 15(4):373–386.
- Chapman, G. B., Nelson, R., and Hier, D. B. (1999). Familiarity and time preferences: Decision making about treatments for migraine headaches and Crohn's disease. *Journal of Experimental Psychology: Applied*, 5(1):17.
- Chapman, G. B. and Winquist, J. R. (1998). The magnitude effect: Temporal discount rates and restaurant tips. *Psychonomic Bulletin & Review*, 5(1):119–123.
- Chesson, H. and Viscusi, W. K. (2000). The heterogeneity of time–risk tradeoffs. *Journal of Behavioral Decision Making*, 13(2):251–258.

- Cheung, S. L. (2016). Recent developments in the experimental elicitation of time preference. *Journal of Behavioral and Experimental Finance*, 11:1–8.
- Coller, M. and Williams, M. B. (1999). Eliciting individual discount rates. *Experimental Economics*, 2(2):107–127.
- Davis, D. D. and Holt, C. A. (1993). *Experimental economics*. Princeton university press.
- Deck, C. and Jahedi, S. (2015a). An experimental investigation of time discounting in strategic settings. *Journal of Behavioral and Experimental Economics*, 54:95– 104.
- Deck, C. and Jahedi, S. (2015b). Time discounting in strategic contests. *Journal of Economics & Management Strategy*, 24(1):151–164.
- Depositario, D. P. T., Nayga, R. M., Wu, X., and Laude, T. P. (2009). Should students be used as subjects in experimental auctions? *Economics Letters*, 102(2):122–124.
- Dolan, P. and Gudex, C. (1995). Time preference, duration and health state valuations. *Health economics*, 4(4):289–299.
- Doucouliagos, C. and Stanley, T. D. (2013). Are all economic facts greatly exaggerated? Theory competition and selectivity. *Journal of Economic Surveys*, 27(2):316–339.
- Dreyfus, M. K. and Viscusi, W. K. (1995). Rates of Time Preference and Consumer

Fuel Efficiency. The Journal of Law and Economics, 38(1):79-105.

- Duquette, E., Higgins, N., and Horowitz, J. (2012). Farmer discount rates: Experimental evidence. In American Journal of Agricultural Economics, volume 94, pages 451-456. Oxford University Press.
- Egger, M., Davey Smith, G., Schneider, M., and Minder, C. (1997). Bias in meta-analysis detected by a simple, graphical test. British Medical Journal, 315(7109):629-34.
- Eicher, T. S., Papageorgiou, C., and Raftery, A. E. (2011). Default priors and predictive performance in Bayesian model averaging, with application to growth determinants. Journal of Applied *Econometrics*, 26(1):30–55.
- Fearnside, P. M. (2002). Time preference in global warming calculations: A proposal for a unified index. Ecological Economics, 41(1):21–31.
- Fernández, C., Ley, E., and Steel, M. F. (2001). Benchmark priors for Bayesian model averaging. Journal of Econometrics, 100(2):381-427.
- Fisher, I. (1930). The theory of interest. New York, 43.
- Foley, D. K., Rezai, A., and Taylor, L. (2013). The social cost of carbon emissions: Seven propositions. Economics Letters, 121(1):90–97.

- Valuations of Automobile Safety and Frank, R. H., Gilovich, T., and Regan, D. T. (1993). Does Studying Economics Inhibit Cooperation? Journal of Economic *Perspectives*, 7(2):159–171.
 - Frederick, S., Loewenstein, G., and O'donoghue, T. (2002). Time Discounting and Time Preference: A Critical Review. Journal of Economic Literature, 40(2):351-401.
 - Fujii, T. and Karp, L. (2008). Numerical analysis of non-constant pure rate of time preference: A model of climate policy. Journal of Environmental Economics and Management, 56(1):83–101.
 - Goulder, L. H. and Stavins, R. N. (2002). Discounting: An eye on the future. Nature, 419(6908):673-674.
 - Hardisty, D. J., Thompson, K. F., Krantz, D. H., and Weber, E. U. (2013). How to measure time preferences: An experimental comparison of three methods. Judgment and Decision Making, 8(3):236-249.
 - Harrison, G. W., Igel Lau, M., Rutström, E. E., and Sullivan, M. B. (2005). Eliciting risk and time preferences using field experiments: Some methodological issues. In Field experiments in economics, pages 125–218. Emerald Group Publishing Limited.
 - Harrison, G. W., Lau, M. I., and Rutström, E. E. (2010). Individual discount rates and smoking: Evidence from a field ex-

economics, 29(5):708-717.

- Harrison, G. W., Lau, M. I., and Williams, M. B. (2002). Estimating individual discount rates in Denmark: A field ex-American economic review, periment. 92(5):1606–1617.
- Havranek, T., Horvath, R., Irsova, Z., and Rusnak, M. (2015a). Cross-country heterogeneity in intertemporal substitution. Journal of International Economics, 96(1):100-118.
- Borders Really Slash Trade? A Meta-Analysis.
- Havranek, T., Irsova, Z., Janda, K., and Zilberman, D. (2015b). Selective reporting and the social cost of carbon. *Energy Economics*, 51:394–406.
- Havranek, T., Irsova, Z., and Vlach, T. (2018a). Measuring the Income Elasticity of Water Demand: The Importance of Publication and Endogeneity Biases. *Land Economics*, 94(2):32.
- Havranek, T., Irsova, Z., and Zeynalova, O. (2018b). Tuition Fees and University Enrolment: A Meta-Regression Analysis. Oxford Bulletin of Economics and Statistics, 80(6):1145-1184.
- Havranek, T., Rusnak, M., and Sokolova, A. (2017). Habit formation in consumption: A meta-analysis. European Economic Review, 95:142-167.

periment in Denmark. Journal of health Hinvest, N. S. and Anderson, I. M. (2010). The effects of real versus hypothetical reward on delay and probability discounting. Quarterly Journal of Experimental Psychology, 63(6):1072–1084.

- Holt, C. A. and Laury, S. K. (2002). Risk Aversion and Incentive Effects. American Economic Review, 92(5):1644-1655.
- Ioannidis, J. P., Stanley, T. D., and Doucouliagos, H. (2017). The Power of Bias in Economics Research. Economic Journal, 127(605):F236-F265.
- Havranek, T. and Irsova, Z. (2017). Do Johnson, M. W. and Bickel, W. K. (2002). Within-subject comparison of real and hypothetical money rewards in delay discounting. Journal of the Experimental Analysis of Behavior, 77(2):129–146.
 - Kass, R. E. and Raftery, A. E. (1995). Bayes factors. Journal of the American Statistical Association, 90(430):773-795.
 - Kühberger, A., Schulte-Mecklenbeck, M., and Perner, J. (2002). Framing decisions: Hypothetical and real. Organizational Behavior and Human Decision Processes, 89(2):1162-1175.
 - Laibson, D. (1997). Golden Eggs and Hyperbolic Discounting. The Quarterly Journal of Economics.
 - Laury, S. K., McInnes, M. M., and Todd Swarthout, J. (2012). Avoiding the curves: Direct elicitation of time preferences. Journal of Risk and Uncertainty, 44(3):181-217.

- Lawrance, E. C. (1991). Poverty and the Rate of Time Preference: Evidence from Panel Data. *Journal of Political Economy*, 99(1):54–77.
- Ley, E. and Steel, M. F. (2009). On the effect of prior assumptions in Bayesian model averaging with applications to growth regression. *Journal of Applied Econometrics*, 24(4):651–674.
- Locey, M. L., Jones, B. A., and Rachlin, H. (2011). Real and hypothetical rewards. *Judgment and decision making*, 6(6):552–564.
- Loewenstein, G. (1987). Anticipation and the Valuation of Delayed Consumption. *The Economic Journal*, 97(387):666.
- Loewenstein, G., Read, D., and Baumeister, R. F. (2003). *Time and decision : economic and psychological perspectives on intertemporal choice*. Russell Sage Foundation.
- Madigan, D., York, J., and Allard, D. (1995). Bayesian Graphical Models for Discrete Data. *International Statistical Review / Revue Internationale de Statistique*, 63(2):215.
- Malesza, M. (2018). The effects of potentially real and hypothetical rewards on effort discounting in a student sample. *Personality and Individual Differences*.
- Marwell, G. and Ames, R. E. (1981). Economists free ride, does anyone else?. Experiments on the provision of public

goods, IV. *Journal of Public Economics*, 15(3):295–310.

- Mazur, J. E. (1984). Tests of an equivalence rule for fixed and variable reinforcer delays. *Journal of Experimental Psychology: Animal Behavior Processes*.
- McClure, S. M., Ericson, K. M., Laibson, D. I., Loewenstein, G., and Cohen, J. D. (2007). Time discounting for primary rewards. *Journal of neuroscience*, 27(21):5796–5804.
- Meier, S. and Sprenger, C. (2010). Presentbiased preferences and credit card borrowing. *American Economic Journal: Applied Economics*, 2(1):193–210.
- Newell, R. G. and Siikamäki, J. (2015). Individual time preferences and energy efficiency. *American Economic Review*, 105(5):196–200.
- Phelps, E. S. and Pollak, R. A. (1968). On Second-Best National Saving and Game-Equilibrium Growth. *The Review* of Economic Studies.
- Raftery, A. E., Madigan, D., and Hoeting, J. A. (1997). Bayesian Model Averaging for Linear Regression Models. *Journal of the American Statistical Association*, 92(437):179–191.
- Read, D. and Read, N. L. (2004). Time discounting over the lifespan. *Organizational behavior and human decision processes*, 94(1):22–32.

- Samuelson, P. (1937). Note on Measurement of Utility. Review of Economic Studies, 4(2):155-161.
- Stanley, T. D. (2005). Beyond Publication Bias. Journal of Economic Surveys, 19(3):309-345.
- Stanley, T. D. (2008). Meta-regression methods for detecting and estimating empirical effects in the presence of publication selection. Oxford Bulletin of Economics and Statistics, 70(1):103-127.
- Stanley, T. D. and Doucouliagos, H. (2010). Picture this: A simple graph that reveals much ado about research. Journal of Economic Surveys, 24(1):170-191.
- Meta-regression approximations to reduce publication selection bias. Research Synthesis Methods, 5(1):60–78.
- Takeuchi, K. (2011). Non-parametric test of time consistency: Present bias and future bias. Games and Economic Behavior, 71(2):456-478.
- Tanaka, T., Camerer, C. F., and Nguyen, Q. (2010). Risk and time preferences: Linking experimental and household survey data from Vietnam. American Economic Review, 100(1):557-571.
- Tol, R. S. J. (1999). Time Discounting and Optimal Emission Reduction: An Ap-

plication of FUND. Climatic Change, 41(3-4):351-362.

- Tsukayama, E. and Duckworth, A. L. (2010). Domain-specific temporal discounting and temptation. Judgment and *Decision Making*, 5(2):72–82.
- Tversky, A. and Kahneman, D. (1981). The framing of decisions and the psychology of choice. Science, 211(4481):453-458.
- Wang, M., Rieger, M. O., and Hens, T. (2016). How time preferences differ: Evidence from 53 countries. Journal of Economic Psychology, 52:115–135.
- Stanley, T. D. and Doucouliagos, H. (2014). Warner, J. T. and Pleeter, S. (2001). The Personal Discount Rate: Evidence from Military Downsizing Programs. American Economic Review, 91(1):33–53.
 - Zauberman, G., Kim, B. K., Malkoc, S. A., and Bettman, J. R. (2009). Discounting time and time discounting: Subjective time perception and intertemporal preferences. Journal of Marketing Research, 46(4):543-556.
 - Zeugner, S. and Feldkircher, M. (2015). Bayesian Model Averaging Employing Fixed and Flexible Priors: The BMS Package for R. Journal of Statistical Software, 68(4):1-37.

Appendices

A PAT-PEESE Robustness Check

PET tests	OLS	FE	Precision	Study
SE (publication bias)	7.993**	7.865**	14.18***	5.640***
	(2.427)	(2.166)	(3.028)	(1.360)
Constant (corrected effect)	0.148^{*}	0.151**	0.0178	0.192^{*}
	(0.0584)	(0.0471)	(0.0192)	(0.0754)
Observations	252	252	252	252
PEESE tests	OLS	FE	Precision	Study
SE^2 (publication bias)	67.65*	88.99**	157.1**	47.95***
	(24.57)	(29.63)	(56.50)	(13.26)
Constant (corrected effect)	0.258**	0.238***	0.0792*	0.260***
	(0.0704)	(0.0280)	(0.0358)	(0.0755)
Observations	252	252	252	252

Table 6: Funnel asymmetry tests for observations with reported SE

Notes: The table reports results of the regressions 5 (PET test) and 6 (PEESE test) for observations with reported SE. All variables winsorized at p = 0.025. The standard errors of the regression parameters are clustered at the study level and shown in parentheses. OLS = ordinary least squares, FE = study-level fixed effects. Precision = weighted by the inverse of the standard error. Study = weighted by the inverse of the number of estimates reported per study. * p < 0.10, ** p < 0.5, *** p < 0.01.

B Supplementary Diagnostics of BMA

Mean no. regressors	Draws	Burn-ins	Time
6.5881	2e+06	1e+06	4.447921 minutes
No. models visited 524963 Corr PMP 0.9999 Shrinkage-Stats Av=0.9977	Modelspace 2097152 No. Obs 434	Visited 25% Model Prior uniform / 10.5	Topmodels 98% g-Prior UIP

Table 7: Summary of BMA estimation

Notes: In this specification we employ the priors suggested by Eicher et al. (2011) based on the predictive performance: the uniform model prior (each model has the same prior probability) and the unit information prior (the prior provides the same amount of information as one observation of the data).

Figure 6: Correlation matrix of BMA variables





Figure 7: Model size and convergence for UIP prior

Notes: Notes: The figure shows Model size and convergence for UIP prior.

C Robustness Checks of BMA

Figure 8: Model inclusion in BMA (based on "BRIC" g-prior)



Model Inclusion Based on Best 5000 Models

Notes: Notes: The response variable is the estimate of discount rate. Columns denote individual models; variables are sorted by posterior inclusion probability in descending order. The horizontal axis denotes cumulative posterior model probabilities; only the 5.000 best models are depicted. Estimation based on "BRIC" Zellner's *g-prior* according to Fernández et al. (2001) and "random" model probability prior suggested by Ley and Steel (2009). Blue color (darker in greyscale) depicts included variables with positive estimated sign. Red color (lighter in greyscale) depicts included variables with negative estimated sign. Variable with no color is not included in the model. Numerical results of BMA exercise are reported in Table 5.

	BRIC g-prior			Without resampling		
Variable	Post. Mean	Post. SD	PIP	Post. Mean	Post. SD	PIP
Intercept	0.248	NA	1.000	0.084	NA	1.000
Standard error combined	1.443	0.219	1.000	NA	NA	NA
Standard error	NA	NA	NA	3.530	0.386	1.000
Estimation characteristics						
disc_factor	-0.143	0.166	0.478	NA	NA	NA
hyperbolic	0.001	0.013	0.025	-0.002	0.015	0.092
exponential	-0.002	0.019	0.029	-0.105	0.074	0.753
ln_timelength	-0.076	0.027	0.945	-0.043	0.012	0.991
idtimehorizon	0.002	0.021	0.034	-0.000	0.011	0.072
method_exp	0.006	0.041	0.046	0.004	0.019	0.098
Experimental characteristics						
reward_real	0.002	0.023	0.031	0.031	0.051	0.361
elicmethod_match	0.006	0.048	0.040	-0.003	0.025	0.128
domain_health	0.333	0.116	0.952	-0.000	0.017	0.063
domain_other	0.751	0.102	1.000	0.684	0.085	1.000
framing_negative	-0.000	0.021	0.019	-0.002	0.024	0.071
framing_neutral	0.001	0.026	0.019	NA	NA	NA
Subject pool characteristics						
ln_samplesize	0.000	0.006	0.019	0.000	0.006	0.067
students	-0.000	0.012	0.022	-0.002	0.016	0.089
male	-0.001	0.025	0.019	0.002	0.020	0.064
female	-0.002	0.029	0.023	0.002	0.021	0.067
continent_namerica	-0.003	0.030	0.031	-0.016	0.040	0.203
continent_asia	0.084	0.137	0.320	0.241	0.072	0.977
Publication characteristics						
ln_google_per_year	0.001	0.009	0.052	0.026	0.019	0.731
std_pubyear	0.004	0.027	0.053	0.004	0.020	0.200
N	434			252		

Table 8: Robustness checks to heterogeneity in discount rate estimates

Notes: We do not include variables disc_factor and framing_neutral into the BMA analysis without resampling of missing standard errors since these variables are not present in reduced dataset. SD = standard deviation, PIP = Posterior inclusion probability, SE = standard error.



Figure 9: Model size and convergence for BRIC prior

Notes: Notes: The figure shows Model size and convergence for BRIC prior.

Figure 10: Model inclusion in BMA with non-combined SE (based on "UIP" g-prior)



Model Inclusion Based on Best 5000 Models

Notes: Notes: Robustness check includes observations with reported measure of statistical precision, i.e. with obtainable standard errors. The response variable is the estimate of discount rate. Columns denote individual models; variables are sorted by posterior inclusion probability in descending order. The horizontal axis denotes cumulative posterior model probabilities; only the 5.000 best models are depicted. Estimation based on "UIP" Zellner's *g-prior* according to Fernández et al. (2001) and "random" model probability prior suggested by Ley and Steel (2009). Blue color (darker in greyscale) depicts included variables with positive estimated sign. Red color (lighter in greyscale) depicts included variables with negative estimated sign. Variable with no color is not included in the model. Numerical results of BMA exercise are reported in Table 5.



Figure 11: Model size and convergence for non-combined SE (based on "UIP" g-prior)

Notes: Notes: The figure shows Model size and convergence for UIP prior. Robustness check includes observations with reported measure of statistical precision, i.e. with obtainable standard errors.

IES Working Paper Series

2018

- 1. Karel Janda, Martin Strobl: *Smoking Czechs: Modeling Tobacco Consumption and Taxation*
- 2. Karel Janda, Michaela Koscova: Photovoltaics and the Slovak Electricity Market
- 3. Simona Malovana, Dominika Kolcunova, Vaclav Broz: *Does Monetary Policy Influence Banks' Perception of Risks?*
- 4. Karolina Vozkova: *Why Did EU Banks Change Their Business Models in Last Years and What Was the Impact of Net Fee and Commission Income on Their Performance?*
- 5. Jan Malek, Lukas Recka, Karel Janda: *Impact of German Energiewende on Transmission Lines in the Central European Region*
- 6. David Svacina: Devaluation with Exchange rate Floor in a Small Open Economy
- 7. Ladislav Kristoufek: *Are the Crude Oil Markets Really Becoming More Efficient over Time? Some New Evidence*
- 8. Karel Janda, Zuzana Lajksnerova, Jakub Mikolasek: *A General Equilibrium Model* of Optimal Alcohol Taxation in the Czech Republic
- 9. Nicholas Tyack, Milan Scasny: *Estimating the Value of Crop Diversity Conservation Services Provided by the Czech National Programme for Agrobiodiversity*
- 10. Laure de Batz: *Financial Impact of Regulatory Sanctions on French Listed Companies*
- 11. Matej Opatrny: *Extent of Irrationality of the Consumer: Combining the Critical Cost Eciency and Houtman Maks Indices*
- 12. Mojmir Hampl, Tomas Havranek: *Foreign Capital and Domestic Productivity in the Czech Republic*
- 13. Miroslav Palansky: *The Value of Political Connections in the Post-Transition Period: Evidence from the Czech Republic*
- 14. Karel Janda: Earnings Stability and Peer Selection for Indirect Valuation
- 15. Ondrej Tobek, Martin Hronec: Does the Source of Fundamental Data Matter?
- 16. Stefan Schmelzer, Michael Miess, Milan Scasny, Vedunka Kopecna: *Modelling Electric Vehicles as an Abatement Technology in a Hybrid CGE Model*
- 17. Barbora Malinska, Jozef Barunik: *Volatility Term Structure Modeling Using Nelson-Siegel Model*
- 18. Lubomir Cingl, Vaclav Korbel: *Underlying Motivations For Rule-Violation Among Juvenile Delinquents: A Lab-in-the-Field Experiment*
- 19. Petr Jansky, Marek Sedivy: *Estimating the Revenue Costs of Tax Treaties in Developing Countries*
- 20. Yao Wang, Zdenek Drabek, Zhengwei Wang: *The Predicting Power of Soft Information on Defaults in the Chinese P2P Lending Market*
- 21. Matej Kuc: Cost Efficiency of European Cooperative Banks

- 22. Dominika Kolcunova, Tomas Havranek: *Estimating the Effective Lower Bound for the Czech National Bank's Policy Rate*
- 23. Petr Jansky, Markus Meinzer, Miroslav Palansky: *Is Panama Really Your Tax Haven? Secrecy Jurisdictions and the Countries They Harm*
- 24. Petr Jansky, Marek Sedivy: *How Do Regional Price Levels Affect Income Inequality? Household-Level Evidence from 21 Countries*
- 25. Mojmir Hampl, Tomas Havranek: *Central Bank Capital as an Instrument of Monetary Policy*
- 26. Petr Pleticha: *Entrepreneurship in the Information Age: An Empirical Analysis of the European Regions*
- 27. Tereza Palanska: *Measurement of Volatility Spillovers and Asymmetric Connectedness on Commodity and Equity Markets*
- 28. Eva Hromadkova, Oldrich Koza, Petr Polak and Nikol Polakova: *The Bank Lending Survey*
- 29. Martin Gregor: Electives Shopping, Grading Competition, and Grading Norms
- 30. Lubos Hanus, Lukas Vacha: *Time-Frequency Response Analysis of Monetary Policy Transmission*
- 31. Matej Opatrny: *The Impact of Agricultural Subsidies on Farm Production: A Synthetic Control Method Approach*
- 32. Karel Janda, Ladislav Kristoufek: *The Relationship Between Fuel, Biofuel and Food Prices: Methods and Outcomes*
- 33. Karel Janda, Jakub Kourilek: Residual Shape Risk on Czech Natural Gas Market
- 34. Martin Stepanek, Kaveh Jahanshahi: *Structural Analysis of Influences on Workplace Productivity Loss*
- 35. Iftekhar Hasan, Roman Horvath, Jan Mares: Finance and Wealth Inequality
- 36. Sarah Godar: *Tax Haven Investors and Corporate Profitability Evidence of Profit Shifting by German-Based Affiliates of Multinational Firms*
- 37. Jan Zacek: Should Monetary Policy Lean against the Wind? An Evidence from a DSGE Model with Occasionally Binding Constraint
- 38. Petr Jansky: *European Banks and Tax Havens: Evidence from Country-by-Country Reporting*
- 39. Mohammad Ali Elminejad: Contagious Defaults in Inter-bank Networks
- 40. Jindrich Matousek: *Individual Discount Rates: A Meta-Analysis of the Experimental Evidence*

All papers can be downloaded at: <u>http://ies.fsv.cuni.cz</u>.



Univerzita Karlova v Praze, Fakulta sociálních věd Institut ekonomických studií [UK FSV – IES] Praha 1, Opletalova 26 E-mail : <u>ies@fsv.cuni.cz</u> http://ies.fsv.cuni.cz